Author’s response to reviews

Title: Neonatal, Infant, and Child Mortality Among Women Exposed to Intimate Partner Violence in East Africa: A Multi-Country Analysis

Authors:

Peter Memiah (pmemiah@ihv.umaryland.edu)
Tristi Bond (tristi.ahmu@gmail.com)
Yvonne Opanga (Yvonne.Opanga@amref.org)
Caroline Kingori (kingori@ohio.edu)
Courtney Cook (ckc18@students.uwf.edu)
Michelle Mwangi (michmwangi18@gmail.com)
Nyawira Gitahi-Kamau (gnyawira@gmail.com)
Deus Bazira (DBazira@ihv.umaryland.edu)
Kevin Owuor (kevinowuor1@gmail.com)

Version: 6 Date: 11 Oct 2019

Author’s response to reviews:

Thank you Editor.

We have run everything from the abstract and the main text through the plagiarism checker and made edits to reduce the overlap to zero (according to the checker).

The list of acronyms is already in text as requested.

We have added in-text citations for Tables 1, 4, and 5.

The tables are now clean